Author’s response to reviews

Title: An unexpected, mild phenotype of glucocorticoid resistance associated with glucocorticoid receptor gene mutation - Case report and review of the literature

Authors:
Ágnes Molnár (agimolnar88@gmail.com)
Attila Patocs (patocs.attila@med.semmelweis-univ.hu)
Gábor Nyiró (nyirogabor1@gmail.com)
Károly Rácz (racz@bel2.sote.hu)
Miklós Tóth (totmik@gmail.com)
Beatrix Sármán (sarmanb@hotmail.com)

Version: 3 Date: 12 Feb 2018

Author’s response to reviews:

Dear dr Pasini,

Please find our revised manuscript entitled "An unexpected, mild phenotype of glucocorticoid resistance associated with glucocorticoid receptor gene mutation - Case report and review of the literature" for consideration to be published in the BMC Medical Genetics.

We thank the editor and reviewers for their comments and their positive feedbacks.

In the revised version we corrected all Technical Comments and all sections have been made according to the journal's requirements.

According to the comments from Reviewer 4 we corrected all typos found in our manuscript and we accepted his suggestions.
We submit a clean version.

We hope that this version will be finally accepted for publication in your prestigious journal.

Best regards

Attila Patocs